Author’s response to reviews

Title: Gastroesophageal reflux symptoms in infants in a rural population: longitudinal data over the first six months

Authors:

Robert S Van Howe (vanhowe@miuplink.net)
Michelle R Storms (mrstorms@mgh.org)

Version: 4 Date: 15 January 2010

Author’s response to reviews:

The details of the comparison of I-GERQ-R scores in those currently primarily breastfed at the various postnatal visits has been included in the text.